 CHOHAN /15-August-05

Week 33  starting 15 Aug 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  15 August 05 : 15/08/2005 15:28:12    
	9
	A1
	MBBL3138
	PT 13.1 IAP @ Warm
	301.89

	1
	A2
	MBAL1084
	HRR @ 1.8K - other
	1.88

	4
	B1
	LE-C11R7
	Thermal cycle - No Tcoils
	5.12 HV

	12
	B2
	MBAL2190
	PT 13.2 Resist. Meas.
	297.68

	10
	C1
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	295.8

	2
	C2
	MBAL2182
	PT 12 Warm Up
	271.82

	3
	D1
	SSS124
	PT 12 Warm Up
	10.41

	11
	D2
	SSS607
	Magnet Alignment
	296.34

	8
	E1
	MBAL3106
	ICS 4 Final connection
	298.04

	6
	E2
	MBAL2208
	ICS 3 Traveller after warm test
	296.34

	7
	F1
	MBBR2220
	ICS 2 Connect Magnet (ICS)
	295.8

	5
	F2
	MBAR3348
	PT 2 HV Cold Test
	1.90


   Tempr. from TBSM18 site
   
Coordinator this week: nick; MTM support 16-3602

14 complete Magnet Tests cycles
Only 10 dipoles ( 0 with MM )+ 1 SSS(without MM & SSW )
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) Last week: Many Thermal cycles (Ansaldo) ; SSS607 (Q4) on D2 long tests; CC on B1 long cooling down…
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
·  Leak on E1  Magnet this week (3106  )
Special  Points :

· Special SSS 607 on D2  (Roy +Marco)
· 1084  Lifetime tests : 200A/s cycles (Giorgio) + Instrumn.

· Connect. Cryostat Tests B1  (Lionel ?) 

Tests 

	Magnets under Test from 07-08-2005 to 13-08-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL2079 
	[image: image3.png]



	


	30-07-2005 
	37 h.
	23 h.
	46 h.
	17 h.
	42 h.
	07-08-2005 
	165 h.
	Stripping
	
	

	2
	TBA1
	MBBL2194 
	[image: image5.png]



	


	07-08-2005 
	46 h.
	22 h.
	15 h.
	14 h.
	10 h.
	11-08-2005 
	107 h.
	Stripping
	
	

	3
	TBA2
	MBBL1190 
	[image: image7.png]



	


	04-08-2005 
	38 h.
	21 h.
	18 h.
	14 h.
	28 h.
	09-08-2005 
	119 h.
	Stripping
	
	

	4
	TBB2
	MBAL1241 
	[image: image9.png]



	


	03-08-2005 
	27 h.
	35 h.
	15 h.
	18 h.
	38 h.
	09-08-2005 
	133 h.
	Stripping
	
	

	5
	TBC1
	MBAL2145 
	[image: image11.png]



	


	07-08-2005 
	35 h.
	11 h.
	17 h.
	19 h.
	16 h.
	11-08-2005 
	98 h.
	Stripping
	
	

	6
	TBC2
	MBAL2182 
	[image: image13.png]



	


	08-08-2005 
	42 h.
	27 h.
	41 h.
	-
	-
	13-08-2005 
	110 h.
	Not Found or Thermal cycle
	
	

	7
	TBC2
	MBAL2184 
	[image: image15.png]



	


	02-08-2005 
	45 h.
	20 h.
	17 h.
	13 h.
	26 h.
	08-08-2005 
	121 h.
	Stripping
	
	

	8
	TBD1
	SSS206 
	[image: image17.png]



	


	02-08-2005 
	16 h.
	23 h.
	58 h.
	65 h.
	0 h.
	12-08-2005 
	162 h.
	Not Found or Thermal cycle
	
	

	9
	TBD1
	SSS206 
	[image: image19.png]



	


	09-08-2005 
	0 h.
	17 h.
	20 h.
	9 h.
	26 h.
	12-08-2005 
	72 h.
	Stand By
	
	

	10
	TBE1
	MBBL2187 
	[image: image21.png]



	


	05-08-2005 
	36 h.
	21 h.
	-
	-
	-
	10-08-2005 
	105 h.
	Not Found or Thermal cycle
	
	

	11
	TBE1
	MBBL2187 
	[image: image23.png]



	


	10-08-2005 
	0 h.
	21 h.
	12 h.
	20 h.
	23 h.
	13-08-2005 
	76 h.
	Stripping
	
	

	12
	TBF1
	MBAL1248 
	[image: image25.png]



	


	05-08-2005 
	23 h.
	23 h.
	17 h.
	14 h.
	15 h.
	09-08-2005 
	92 h.
	Stripping
	
	

	13
	TBF2
	MBBL2166 
	[image: image27.png]



	


	08-08-2005 
	35 h.
	21 h.
	15 h.
	16 h.
	8 h.
	12-08-2005 
	95 h.
	Stripping
	
	

	14
	TBF2
	MBBR3387 
	[image: image29.png]



	


	02-08-2005 
	85 h.
	-
	-
	-
	-
	08-08-2005 
	128 h.
	Stand By
	
	


	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2079
	11970.3 A.
(D2-L)
	12669.3 A.
(D2-L)
	12787 A.
(D1-L)
	[image: image31.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL2194
	11380 A.
(D1_L)
	12311.1 A.
(D2-L)
	12245 A.
(D2_L)
	12444 A.
(D1_U)
	[image: image32.png]



	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL1190
	12256.3 A.
(D1_L)
	12849 A.
(No Data)
	12851 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL1241
	12386.3 A.
(D1-L)
	12849 A.
(D2-L)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL2145
	11757 A.
(D2_L)
	12672.1 A.
(D1-L)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL2182
	10762 A.
(D1-L)
	11231 A.
(D1-L)
	11920 A.
(D1_L)
	12206 A.
(D1_L)
	12303 A.
(D2_U)
	12444 A.
(D1-U)
	12337 A.
(D1_L)
	12397 A.
(D1_U)
	12740 A.
(D1_U)
	[image: image36.png]



	-
	-
	-

	7
	MBAL2184
	9920.7 A.
(D1_L)
	11083.4 A.
(D1_U)
	12266 A.
(D1_U)
	12758.8 A.
(D1_U)
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	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS206
	11323 A.
(QE-34)
	11519 A.
(QE-34)
	11845 A.
(QE-34)
	12001 A.
(QE_12)
	12415.5 A.
(QE-34)
	12420 A.
(QE-12)
	12585 A.
(QI_12)
	12670 A.
(QE-34)
	12795.4 A.
(Q!-34)
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	-
	-
	-

	9
	SSS206
	11858 A.
(E_34)
	12405.9 A.
(QE-34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL2187
	11265 A.
(D1-U)
	11692 A.
(D2-U)
	12201 A.
(D2-U)
	12373 A.
(D2-U)
	12429.9 A.
(D1-U)
	12488 A.
(D1_U)
	12578.4 A.
(D1_L)
	12653.9 A.
(D1-L)
	12784.6 A.
(D1-U)
	[image: image40.jpg]



	-
	-
	-

	11
	MBBL2187
	11924 A.
(D1-U)
	12214.1 A.
(D1_U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAL1248
	12100 A.
(D1_U)
	12387 A.
(D2_L)
	12820 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBL2166
	11927 A.
(D1-U)
	12535 A.
(D2-U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBR3387
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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LEAK on 3387;   2187 had Thermal Cycle ;  sss206  had Thermal cycle ;    2182  had a Thermal cycle 

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday _10_A
	TBD1


	SSS206: Training quench 2, the signals QE, QI & Vtotal were completely flat.
	AB/CO
	
	Giorgio informed. He checked with lfoff and found that one acquisition card is to be replaced: done.

	Wednesday _10_N
	TBD1


	SSS206: PT10.1 (quench provoked at nominal) QH 1-2 discharged without any trigger generated and no QH 3-4 discharged.
	MTM
	
	QH re-charged. Warm_up quench performed. Meantime SPA occurred and no quench detected.

	Thursday_11_N
	TBD1


	SSS206: Resistance checks indicated the MO connection is crossed. No fault was found in the 904
	ICS
	
	SSS put in stand-by.

Error in cover flange. F.Gavin will correct it.

	Friday_12_N
	TBF1


	No communication between PLCSE and PLCHV.

Problem on connector at cable side or HV Rack side
	AB/CO-MTM
	
	Solved temporarily holding up the cable.



	Friday_12_N
	TBD1


	Trigger in V_clamp E on SSS124
	ACR
	
	Giorgio contacted. He advised to ask ICS  to check the connection



	Friday_12_N
	TBA2


	Quench 4 on MB1084 : alarm that Vinit<810V for QH1(802V).
	MTM
	
	Check on QH graph: discharge OK.

It was observed that the quench heater power supply was fluctuating between 802V and 810V. But that was quite stable after some time at 818 volts.

Proceed with tests: QH OK for quench 5.

	Saturday_13_M

Saturday_13_A
	TBB2


	MBAL2190 : AQA alert in PT5.2.
	MTM
	
	Giorgio contacted and he asked to repeat the test. Same alarm. Giorgio asked to perform :

-to conduct resistance measurement 

-HV test (QH Investigation)
Results:

(1) Dipole vs Quench Heater YT222 (525 V): Capacitance is not within expectation.
(2) Quench Heater YT211 vs Ground (1850 V): Current higher than maximum specified value.
(3) Quench Heater YT222 vs Ground (600 V & 1850 V): capacitance is not within expectation.

Discharge checked by Giorgio: YT 211 is not working properly. 

	Saturday_13_A
	TBA2


	MB1084: Magnet reached the ultimate at quench number 9. During the plateau at 12850 A, LF HF pressed to trigger the u file. Magnet quenched @ 12850 before LF could arm.
	MTM
	
	Magnet probably undergo to a thermal cycle after SSL @ 4.2 K.

	Saturday_13_A
	TBB2


	MB2190 : CDAP HV test failed on Dipole-QH vs Ground
	MTM
	
	TC and Giorgio decide d to warm up the magnet with quench at 3 kA with special protection (YT 211 replaced by YT212).

	Sunday_14_A

Sunday _14_N
	TBA2


	Current cycling on MB1084 could not be continued because current lead temperature is 450 deg K.
	ACR
	
	Investigation by Cryogenic Team

	Sunday_14_N
	TBD1


	SSS124: could not power  ‘EF circuit’
	MTM
	
	Giorgio asked to continue with powering of GH and IJ circuits and with training quenches.

	Sunday_14_N
	TBD1


	SSS124: Magnet warm to 2.31 K suddenly
	ACR
	
	One communication card was changed by ACR team.

	Monday 15_A
	TBA1


	IAP at warm final on MB3138: Resistance measurements of the sextupole corrector show an open circuit.
	ACR
	
	Guy informed. 

	Monday 15_A
	TBE1


	Leak on the MRB for MB3106
	ICS
	
	Magnet has to be removed.

	Monday 15_A
	TBF1


	Communication failure between PLCHV and PLCSE. The problem is with the PLCSE cable.
	AB/CO-MTM
	
	Solved temporarily by tying the cable to higher point.


Green Connect + Pump down /Blue :  Wait + Cool down	Ivory: Tests at Cold (36 hr)Red :  Wait + Warm up (12 hr)	  Dark Green: Disconnect (8 hr)











